Author's response to reviews

Title: A unique strain of community-acquired Clostridium difficile in severe complicated infection and death of a young adult.

Authors:

Orville D Heslop (orvilleheslop@gmail.com)
Karen Roye-Green (kjrgreen@yahoo.com)
Kathleen Coard (kathleen.coard@uwimon.com)
Michael R Mulvey (Michael_mulvey@phac-aspc.gc.ca)

Version: 8 Date: 13 June 2013

Author's response to reviews: see over
June 2013

Response to Reviewers (Scott Curry & Liang Chen):

Article title: A unique strain of community-acquired Clostridium difficile in severe complicated infection and death of a young adult.

MS ID: 6664532078909091

Comments/suggestions.

a). The term "hypervirulent" in the Figure for strain ATCC 43255 was deleted as suggested.

b). Adjustments and modifications were made to the abstract to clarity the salient points of the case.

c). Particular attention was paid to grammar and sentence construction, aided by someone who is proficient in the English language and project writing.